Author's response to reviews

Title: Translocation pathways for inhaled asbestos fibers

Authors:

Giuseppe A Miserocchi (giuseppe.miserocchi@unimib.it)
Giulio A Sancini (giulio.sancini@unimib.it)
Francesco Mantegazza (francesco.mantegazza@unimib.it)
Gerolamo Chiappino (fibre@unimi.it)

Version: 5 Date: 15 January 2008

Author's response to reviews:

Point-by-point description of the changes made

Pag. 1, title: removed all unnecessary capitals, removed “Review
Pag. 1, lines 7-11: the list of authors is now exactly as in the submission system
Pag. 1, line 10: removed underlined text
Pag. 2, line 1: removed all unnecessary capitals
Pag. 3, line 1: removed all unnecessary capitals
Pag. 13, line 9: removed all unnecessary capitals
Pag. 14, lines 7, 9, 16: removed all unnecessary capitals
Pag. 14, line 15: added a sentence
Pag. 15, line 1: removed all unnecessary capitals
Pag. 21, line 1: removed all unnecessary capitals

List of references: Ref. 5 e 19 removed “et al.” and all the editors had been listed

Figures were cropped as closely as possible.